



Agius, M. R., Harmon, N., Rychert, C. A., Tharimena, S., & Kendall, J. M. (2018). Sediment Characterization at the Equatorial Mid-Atlantic Ridge From P-to-S Teleseismic Phase Conversions Recorded on the PI-LAB Experiment. *Geophysical Research Letters*, *45*(22), 12244-12252. https://doi.org/10.1029/2018GL080565

Publisher's PDF, also known as Version of record

License (if available): CC BY

Link to published version (if available): 10.1029/2018GL080565

Link to publication record in Explore Bristol Research PDF-document

This is the final published version of the article (version of record). It first appeared online via AGU at https://agupubs.onlinelibrary.wiley.com/doi/full/10.1029/2018GL080565 . Please refer to any applicable terms of use of the publisher.

University of Bristol - Explore Bristol Research General rights

This document is made available in accordance with publisher policies. Please cite only the published version using the reference above. Full terms of use are available: http://www.bristol.ac.uk/pure/about/ebr-terms



Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL080565

Key Points:

- P-to-S delay times from the sediment-crust boundary suggest
 5-82 m of sediment beneath PI-LAB
- The thickness generally agrees with global estimates but is less than expected on older lithosphere
- Higher deposition rates at < 10 Myr suggest higher productivity perhaps related to stronger monsoons

Supporting Information:

Supporting Information S1

Correspondence to:

M. Agius, matthew.agius@soton.ac.uk

Citation:

Agius, M. R., Harmon, N., Rychert, C. A., Tharimena, S., & Kendall, J. M. M. (2018). Sediment characterization at the equatorial Mid-Atlantic Ridge from P-to-S teleseismic phase conversions recorded on the PI-LAB experiment. *Geophysical Research Letters*, *45*. https://doi.org/10.1029/2018GL080565

Received 25 SEP 2018 Accepted 2 NOV 2018 Accepted article online 9 NOV 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Sediment Characterization at the Equatorial Mid-Atlantic Ridge From *P*-to-*S* Teleseismic Phase Conversions Recorded on the PI-LAB Experiment

M. R. Agius^{1,2}, N. Harmon¹, C. A. Rychert¹, S. Tharimena^{1,3}, and J.-M. Kendall⁴

¹Ocean and Earth Science, University of Southampton, Southampton, UK, ²Now at the Department of Geosciences, Faculty of Science, University of Malta, Msida, Malta, ³Now at Jet Propulsion Laboratory, California Institute of Technology, Pasadena, CA, USA, ⁴School of Earth Sciences, University of Bristol, Bristol, UK

Abstract Accurate marine sediment characteristics, for example, thickness and seismic velocity, are important for constraining sedimentation rates with implications for climate variations and for seismic imaging of deeper structures using ocean bottom seismic deployments. We analyze *P*-to-*S* seismic phase conversions from the sediment-crust boundary recorded by the Passive Imaging of the Lithosphere-Asthenosphere Boundary (PI-LAB) experiment to infer the sediment thickness across the Mid-Atlantic Ridge covering 0- to 80-Myr-old seafloor. We find *P*_d*s*-*P* delay times of 0.04–0.37 s, or 5- to 82-m thickness. Sediment thickness increases with age. Thickness agrees with global estimates for young (<15–20 Myr) seafloor but is thinner on older lithosphere. Our result may represent a lower limit on sediment thickness, given that several of our stations are on topographic highs. The sedimentation rate decrease observed from 5 to 1.2 mm/kyr at ~10 Myr suggests a recent increase in productivity related to climate change, eolian dust fluxes, and/or biogenic marine activity.

Plain Language Summary Studies on marine sediment production rate provide historical information on the past climates and sedimentation processes. Unfortunately, such data from the seafloor is sporadic across the oceans, limited to localized drilled sites and seismic profiles. Here we analyze the sediment characteristics along the equatorial Mid-Atlantic Ocean, starting from age 0 at the ridge to sediments 80 million years old. We use newly acquired seismic data from the recent state-of-the-art Passive Imaging of the Lithosphere-Asthenosphere Boundary experiment. Thirty-nine seismometers were put at the ocean bottom down to about 5,000 m below sea level and left recording seismic data for one year between 2016 and 2017. Conversion of seismic phases from the boundary between the crust and the sediment were analyzed to determine the sediment seismic velocity and thickness below each station. It was found that the thickness increases with age at a slower rate than previously thought and that an abrupt increase in the sedimentation rate occurred in the last 10 million years, from 1.2 to 5 mm per thousand years. The recent increase in productivity may be due to factors such as changes in climate, changes in the eolian dust fluxes from Africa to the Atlantic, and/or changes in biogenic marine activity.

1. Introduction

In the deep ocean basins away from continental shelves, sedimentation is composed of biogenic matter, which is a function of local productivity, a terrestrial component due to aerosols and local submarine weathering and erosion. Aerosol and biogenic productivity are a function of the climate and ocean temperature through time and so are important for our understanding of the paleoclimate and oceans. In addition, accurate sediment characteristics (e.g., thickness and seismic velocity) are important for accurately imaging the oceanic crust and mantle. Our understanding of these sediments and sedimentation rates are based on the analysis of core samples extracted from drilling into the ocean seafloor (e.g., Ocean Drilling Program), active source seismic studies (e.g., Mehouachi & Singh, 2018; Seher et al., 2010), and passive source seismic studies (e.g., Harmon et al., 2007; Lewis & Dorman, 1998; Shearer & Orcutt, 1987; Ruan et al., 2014; Rychert et al., 2018). Some of these data are also used to compile a global model of sediment thickness (Whittaker et al., 2013), which incorporates crustal age and distance from the continents. Drilling and active seismic sources provide high resolution but are usually limited in areal extent. Passive seismic deployments offer the opportunity to constrain sediment thickness over a larger area and range of seafloor age.

<mark>_</mark>





Figure 1. *Ps* sediment delay time across the Mid-Atlantic Ridge (MAR). Circles are average measurements, squares are linear interpolated measurements from the two nearest neighbors along the same SW-NE profile, and triangles are stations without data, where no interpolation is applied; together they make the Passive Imaging of the Lithosphere-Asthenosphere Boundary seismic network. Black hexagons: International Ocean Drilling Project (IODP) drilled holes Leg 108 sites 662 and 663 (Ruddiman & Janecek, 1989). Thick gray line indicates the ridge (Bird, 2003). Thin contours: Seafloor age at 20-Myr intervals (Müller et al., 2008). Inset maps: Location of the study area (red box) on a regional map and the earthquakes used in this study (blue dots).

The Passive Imaging of the Lithosphere-Asthenosphere Boundary (PI-LAB) experiment presents a unique opportunity to examine deep ocean basin scale sedimentation. There has been some drilling along the Mid-Atlantic Ridge (MAR; e.g., Karson et al., 1997; Kelemen et al., 2007) with a few sites close to the equator (Figure 1; Ruddiman & Janecek, 1989), although overall the region is sparsely sampled. PI-LAB network consisted of 39 ocean bottom seismometers deployed at the equatorial Mid-Atlantic Ocean on 0- to 80-Myr seafloor from March 2016 to March 2017. The stations were broadband 120-s period (station names starting with letter *L*) and 240-s period (S and I), deployed on both sides of the ridge and centered on the Chain Fracture Zone (Figure 1). Here we use the *P*-to-*S* seismic phase conversions from the sediment-crust boundary to estimate the sediment velocity and thickness beneath each station and infer the sedimentation history of the region.

2. Thickness Estimation

Sediments have a sharp seismic velocity and density contrast to the underlying basaltic crust. Upward propagating compressional (*P*) waves are converted at the boundary between the sediment and the crust to shear (*S*) waves (P_ds), which are both recorded on the ocean bottom seismometers (Figure 2a). The delay time (*dt*) between the two phases (P_ds -*P*) and their relative amplitude depend on thickness and seismic velocities of the sedimentary layer. Here we characterize sediments using teleseismic earthquakes recorded by the PI-LAB experiment, with analysis similar to works by Harmon et al. (2007) and Rychert et al. (2018).

We processed the seismic records in the following way. First, we removed the instrument response. We then determined the station horizontal orientations using *P* wave polarization (Magotra et al., 1989) and Rayleigh wave polarization (Doran & Laske, 2017). Both methods gave horizontal orientations that were within error of each other. The seismic records of teleseismic events with near-vertical raypaths were rotated into the radial and transverse components using the orientation from *P* wave. We demeaned the data, and band-pass filtered between 0.04 and 0.25 Hz. Teleseismic earthquakes that exhibited clear *P* wave arrival on the vertical component and clear *P*-to-*S* converted phase from the sediment-crust interface on the radial component





Figure 2. Examples of *Ps* sediment conversions, and relationship of P_ds/P amplitude ratio and delay time (P_ds-P , dt) with respect to slowness. (a) Schematic of the raypaths of the *P* wave and *P*-to-*S* conversion from the base of the sediment layer, black and red arrows, respectively. (b) Black and red waveforms show the direct *P* wave and the converted *S* wave on the vertical and radial component, respectively. Note that there is no incoming signal on the transverse component (blue seismogram). The maximum absolute amplitude of the *P* phase and the following *S* phase are marked with dots. The seafloor age and delay time are indicated in the bottom left and right, respectively. Each frame is the record from a different station for the same teleseismic earthquake. (c) Top: P_ds/P amplitude ratio with respect to slowness for earthquakes recorded at station S03D. Gray line: P_ds/P ratio from synthetic waveforms using a 1-D model and a range of ray parameters. Bottom: Delay time with respect to slowness for earthquakes recorded at station S03D. Horizontal gray line: Average *Ps* delay time for the station. Note that *dt* remains stable for a wide range of ray parameters. OBS = ocean bottom seismometer.

were chosen by visual inspection. A window one wavelength long from the *P* wave arrival is selected. The timing of the maximum absolute amplitude of the *P* phase and the following P_ds phase peak were then determined automatically within the window. Figure 2b shows examples of *P* and P_ds converted phases at different stations for the same earthquake (24 November 2016, 18:43 UTC, Mw 7.0). The delay time between the two phases is less than 0.5 s, and measurements are robust independent of earthquake magnitude or distance (Figure 2c, bottom frame).

The filtering is necessary to avoid picking secondary peaks in the waveforms due to noise in the data; however, special attention is taken to ensure that delay times, and in turn the sediment thickness, are not underestimated due to the choice of filter (e.g., Ritsema et al., 2009). The delay time could be underestimated at low frequency if the *P* wave and the *P*-to-*S* conversion interfere, which can occur for nonvertical incidence. We do not expect large degrees of interference because the incidence angles of *P* waves in slow sediment is small, typically < 8°. Nevertheless, we also performed a test in which we analyzed the data using a high-pass filter 0.04-20 Hz. There is no systematic bias and the results are in good agreement regardless of filter (supporting information Figure S1). In another test, we determine the accuracy of the sediment delay time from the converted phase given a realistic crustal structure (supporting information Figure S2). We find that difference between the predicted and the measured delay times is small, well within errors. The reason that the filter does not affect our results is because the sediment is thin, and very slow, and the waves are traveling nearly vertically so there is no interference between the *P* and the *P*-to-*S* conversion.

The sediment thickness is inferred using previously determined relationships between delay time and seismic velocities of compressional waves (V_p ; e.g., Nafe & Drake, 1957) and shear waves (V_s ; e.g., Ruan et al., 2014). The empirical function of Ruan et al. (2014), is given by

$$V_{\rm S}(h) = (ah^2 + bh + cV_0)/(h + c)$$

where V_0 is the wave speed of the sediment at the seafloor, 0.1 km/s (e.g., Hamilton, 1979), *h* is the depth below the seafloor, and *a*, *b*, and *c* are three constant parameters 0.15608, 1.2198, and 0.49473, respectively (Bell et al., 2015). These parameters were established by inverting Rayleigh wave ratios of vertical displacement-to-pressure for sediment thicknesses and shear velocity at the seafloor of the Juan de Fuca Ridge (Bell et al., 2015; Ruan et al., 2014).

Geophysical Research Letters



Figure 3. Comparison of estimated sediment thickness with global sediment model, age, subsidence, and production rate. (a) Estimated sediment thickness overlaid on the global sediment model of the world's oceans (Whittaker et al., 2013). Symbols are the same as in Figure 1. (b) Comparison of *P*-to-*S* delay sediment thickness with the global model. Error bars show the uncertainty. Inset: Comparison of estimated sediment thickness with the global model for Cascadia (Rychert et al., 2018). Gray line: One-to-one relationship between the estimated and global model. Different colors represent the seafloor age below the station. Gray dots: Stations without an age constraint. (c) Difference in sediment thickness between estimated and global model (vertical bars), and regional subsidence determined by binning and averaging the acquired bathymetry data (Harmon et al., 2018; *depth* = $2861 + 324.3 \times \sqrt{age}$). (d) Sediment deposition rate, thickness, and seafloor age. Thick gray line: Estimated accumulated sediment thickness over time at the indicated deposition rates. Thin gray line: Inferred compacted sediment thickness (Alibés et al., 1996). Gray background: The depth at each station. Pentagon and circles: Stations located on west and east side of the ridge, respectively.

The empirical relationship of Nafe and Drake (1957) for deep water is given by

$$V_P = 0.43h + 1.83$$

The thickness is calculated from the delay time via the following equation:

$$dt = h(\sqrt{(1/V_S)^2 - u^2} - \sqrt{(1/V_P)^2 - u^2})$$

where *u* is the ray parameter.

We verify this relationship is valid at one example station, S03D, which recorded a large number of waveforms. We compared the observed amplitudes and delay times of the *P*-to-*S* conversions with those from synthetic seismograms (Shearer & Orcutt, 1987) computed assuming the predicted sediment shear velocity (V_s =0.14 km/s) from the previously determined relationship developed for pelagic sediment in Cascadia using compliance (Ruan et al., 2014), and V_p =1.84 km/s from reflection (Nafe & Drake, 1957; Figure 2). We assumed crustal *P* wave velocities increase from 4.5 to 7.5 km/s from the surface to the Moho adapted from seismic refraction measurements in the north MAR (Seher et al., 2010). The predicted amplitudes show good agreement with our observations (Figure 2c).

We further tested this assumption by determining our own best fitting parameters (sediment V_P , sediment V_S , and sediment thickness) at example station S03D. We compared the amplitude and delay time of synthetic seismograms to our observed values using a grid search approach searching over V_P and V_S (supporting information Figure S3). We find that the values of Ruan et al. (2014) and Nafe and Drake (1957), used here, are within 1σ confidence of the grid search. Similar range of V_S values have been obtained from in situ measurements (Hamilton, 1976) and nonlinear waveform fitting (Nolet & Dorman, 1996; V_S =0.31–0.37 km/s at 100-m



Table 1

PI-LAB Stations Longitude, Latitude, Elevation, Average Ps Sediment Delay Time, Number of Events Used, Standard Error, Average Sediment Thickness, and Error

Station	Longitude	Latitude	Elevation (m)	Average delay (s)	NoE	Error (s)	Average thickness (m)	Error (m)
101D	-17.8855	1.2734	-4,047					
L02A	-17.4085	-1.1667	-3,499					
S03D	-17.0315	-2.4021	-3,750	0.23	8	0.06	36	13.61
104D	-16.1733	-2.1238	-3,928	0.29	3	0.03	49	7.27
L05A	-15.4100	-1.8600	-4,052	0.24 ^a			38 ^a	
S06D	-14.4298	-1.6703	-3,778	0.17	6	0.06	25	12.01
107D	-14.0428	-1.5565	-3,819	0.14 ^a			20 ^a	
L08D	-13.6409	-1.4493	-3,357	0.10 ^a			14 ^a	
L09A	-13.3185	-1.3569	-3,378	0.07 ^a			9 ^a	
S10D	-12.9697	-1.3180	-3,015	0.04	4	0.04	5	4.35
S11D	-12.4602	-1.1691	-2,905	0.11	5	0.06	15	9.04
I12D	-10.7766	-0.8683	-4,022	0.25	2	0.03	39	7.84
L13D	-9.5619	-0.5862	-4,659	0.28	2	0.11	50	30.03
I14D	-7.9524	-0.3522	-4,702	0.28	3	0.12	52	29.28
S15D	-6.6228	0.1814	-4,927					
L16D	-7.8953	0.8933	-4,581	0.33	6	0.07	63	25.50
\$17D	-8.5121	0.7422	-5,205	0.31 ^a			56 ^a	
L18D	-9.3765	0.5769	-4,890	0.28	5	0.05	47	12.28
S19D	-9.9754	0.4809	-4,607	0.28 ^a			49 ^a	
120D	-10.5352	0.3681	-4,724	0.29 ^a			51 ^a	
L21D	-11.0380	0.2364	-4,625	0.30	1		52	
S22D	-11.6799	0.1254	-4,352	0.30	11	0.12	58	31.84
L23D	-12.1478	0.0521	-4,631	0.26	2	0.08	43	20.70
S24D	-12.7806	-0.1383	-4,453	0.25 ^a			40 ^a	
L25D	-13.2230	-0.1745	-4,207	0.24 ^a			37 ^a	
S26D	-13.6260	-0.3434	-4,216	0.23	2	0.01	35	3.05
L27A	-13.9427	-0.4007	-3,928	0.21 ^a			32 ^a	
I28D	-14.2684	-0.4918	-3,711	0.20	3	0.05	29	8.59
S29D	-14.6272	-0.5597	-3,626	0.23	3	0.10	37	24.20
L30A	-14.9467	-0.5880	-4,003	0.17	4	0.09	25	18.32
S31D	-15.3187	-0.7141	-3,408	0.18	5	0.11	28	22.10
\$32D	-15.6470	-0.7968	-2,967	0.12 ^a			18 ^a	
L33D	-16.0152	-0.8747	-3,933	0.06	3	0.02	7	2.38
I34D	-16.3485	-0.9579	-2,964	0.10	2	0.08	14	11.24
S35D	-16.6798	-1.0372	-3,773					
136D	-17.0306	-1.1170	-3,938					
L37D	-14.9718	1.5657	-5,054					
S38D	-12.7623	1.9218	-4,926	0.37	6	0.09	82	34.93
L39D	-11.4904	2.0557	-4,685					

Note. PI-LAB = Passive Imaging of the Lithosphere-Asthenosphere Boundary.

^aLinear interpolated values. NoE: Number of events.

thick sediments). Lower values for V_{ρ} (~1.55 km/s) have been obtained from nearby cores drilled by the International Ocean Drilling Project (IODP sites 662 and 663; Figure 1; Shipboard Scientific Party, 1988a, 1988b). However, assuming this V_{ρ} results in minimal changes in sediment thickness (supporting information Figure S3). Similarly, other relationships for V_{ρ} thickness relationships would have little effect on our results, given the similarity of many other relationships to that of Nafe and Drake (1957; e.g., Hamilton, 1979) and also that we have little sensitivity to V_{ρ} velocity-thickness relationships (Ruan et al., 2014). Of course thickness and velocity trade-off. However, our tests at S03D suggest that the previously determined velocity-thickness relationships are a good match. In addition, the gird search test at S03D suggests that, if anything, the thickness could be lower than that determined using the previously derived relationships. We did not record a sufficient number of usable waveforms to allow us to determine our own sediment thickness-shear wave velocity relationship at the other stations in the array because the uncertainty is too high.

In total, we measured 86 *P*-to-*S* conversions from 16 earthquakes with magnitude > 6.0 Mw, recorded on 21 stations. Sediment properties are interpolated to station locations that did not yield a reliable result so that they may be better accounted for in future seismic imaging work. The linear interpolation is between the two adjacent stations along the same SW-NE profiles (Figures 1 and 3a). We did not interpolate to stations that did not record data or were not recovered. The average delay times, sediment thicknesses, standard errors, and interpolated values of all the stations are listed in Table 1 and shown in Figure 3. The error in our estimated sediment thickness is relatively large in comparison to the thicknesses themselves, sometimes over 50%. However, this is expected given that thicknesses are very small and also the frequency content of our waveforms. Despite this error, we observe increasing thickness with age, as expected, and a clear age-thickness trend in the data, again suggesting that the result is robust.

3. Sediment Thickness Along the Equatorial MAR

The analysis of *P* and P_ds sediment phases using the PI-LAB network gives a broad coverage over more than 1,000 km across the equatorial MAR, from 0.3 Myr old to 70.9 Myr (Figure 1). The delay time between the two phases shows a clear progression with depth and age starting with an average of 0.04 s at the ridge increasing to an average of 0.37 s (Table 1). The migrated sediment thicknesses are in the range of 5–82 m and increase with age and depth (Figure 3).

In another approach, we confirmed the thickness of the sediments by analyzing the subbottom profiler data recorded from the onboard Knudsen echo sounder. In principle the sounder is used to determine the depth of the seafloor; however, subsurface reflections from sedimentary layers are sometimes visible (supporting information Figure S4). For example, the two-way travel time profile for station L13D shows a sediment delay time of about 0.035 s between the upper and lower sediment interfaces yielding a thickness of about 54–64 m for V_p of 1.55–1.84 km/s, respectively, within the error of the *P*-to-*S* delay sediment thickness (50 \pm 30 m, Table 1). Interestingly, the thicknesses determined from these two different, independent techniques are both smaller than the global model. Unfortunately, we only had Knudsen data available at a few sites to do this kind of comparison.

Our results agree well with the global total-sediment-thickness model of Whittaker et al. (2013) up to about 15-20 Myr, after which, our sediment thicknesses are significantly less than the global model (Figures 3a-3c). A similar *P*-to-*S* study of sediment thickness on the 0- to 10-Myr-old Juan de Fuca and Gorda Plates (Rychert et al., 2018) also found a one-to-one trend with the global model except for beneath the continental margin, where *P*-to-*S* thicknesses were greater than the global model, but in agreement with refraction work (Horning et al., 2016; Figure 3b, inset chart).

In general, the sediment thickness increases with age but our results show a clear change in slope, steeper < 10 Myr closer to the ridge, and less steep on the older seafloor (Figure 3d). The slopes are a long-term proxy for sedimentation rate. We determine a sediment deposition rate that fits the data, taking into consideration the compaction of sediments over time. We adopt the relationship for decompaction determined for the Madeira Abyssal Plain in the North Atlantic by Alibés et al. (1996). They studied various lithologic layers using data acquired from seismic reflection lines and borehole data and found a compaction of approximately 40% of the original thickness, represented by

decompacted depth = $0.6974 \times compacted depth^{(1.1507)}$

(Alibés et al., 1996; Alibés et al., 1999). We find our compacted sediment thickness can be fit with a sediment deposition rate of 5 mm/kyr for young seafloor (< 10 Myr), and a deposition rate of 1.2 mm/kyr for seafloor older than 10 Myr (Figure 3d). We also find that sediment thickness is symmetric across both sides of the ridge, with the two sedimentation rates applicable to either side, at least up to 25 Myr (Figure 3d). This means that both flanks have undergone the same sedimentation process.

Our results can be compared with direct sediment analysis from IODP cores drilled on the eastern side of the ridge in seafloor 3.7 Myr old, close to our southern profile (Figure 1), and box cores throughout the region (Ku et al., 1968). The IODP cores sampled sediments to 200 meters depth (Ruddiman & Janecek, 1989) much deeper than our 15-m result from nearby station S11D, which is about 65 and 84 km away. The very thick sediments were attributed to the fact that the sites were located in local basins close to the ridge, in an area characterized by strong topographic variations in the range of $\sim 1-3$ km at very short distances < 10-20 km (Harmon et al., 2018), receiving more sediment than the surrounding topographic highs with several slump/turbidity current events visible in more than 50% of the section (Ruddiman & Janecek, 1989). They propose a local deposition rate between 30 and 50 mm/kyr, which included the additional depositional events. Our deposition rate of 1.2 mm/kyr, however, match closely with rates derived from the ionium age from box cores sites close to our study region (0.8-1.5 mm/kyr; Ku et al., 1968).

Agreement with the global model of sediment thickness up to 15-20 Myr suggests that the large increase in sedimentation rate at 10 Myr is likely encompassed in the global model. Our thinner sediments on seafloor older than 15-20 Myr could be at least in part related to the selection of the sites which were targeted for local topographic highs (Figure 3c). Thirteen out of 21 stations are located on shallower bathymetry than predicted by half-space cooling (Figure 3c). This means that our result probably represents a lower bound on sediment thickness and rate. An extreme example of this is S38D, which has a thickness of 82 m located on seafloor 70.9 Myr, much thinner than the 486-m thickness found in the global model. The thin sediment at this location is also confirmed from the subbottom profiles (supporting information Figure S4). Such thin sediments are not uncommon and can be found elsewhere across the Atlantic (e.g., Ewing & Ewing, 1967; Mehouachi & Singh, 2018). This station is located on a regional topographic high, whereas the global model is likely influenced by terrestrial sediments coming off of Africa. Overall agreement of station S38D with the sediment deposition rates and trends found at all of our stations, and disagreement with the global model, suggests that our sediment deposition rates represents an upper limit, appropriate for topographic highs, in comparison to the global model which represents an upper limit appropriate for basins.

The observed change in the rate of deposition at 10 Myr is likely related to several important climatic changes that occurred 7–10 Ma, with even greater enhancement in the past 3 Ma, caused by increased glaciation (Ravelo et al., 2004). Specifically, data from marine sediment cores suggest that eolian dust fluxes from Africa to the Atlantic varied dramatically during the last 10 Myr (e.g., Ruddiman, Sarnthein, et al., 1989; Tiedemann et al., 1994). One contributing factor for the change in dust flux is tectonically driven large-scale, rapid uplift during the latest Cenozoic (8–10 Myr) in East Africa, and South America (Ruddiman, Prell, et al., 1989) and farther away in Southeast Asia (Molnar et al., 1993), which affect climate patterns such as the African and Asian Monsoon. In addition, the shrinkage of the Tethys Sea during the Tortonian (7–11 Myr) could have affected the African summer monsoon (Zhang et al., 2014). In the last 3 Myr there has been a decrease in the North Atlantic sea surface temperatures (de Menocal, 1995), related to the gradual global cooling in the Pliocene epoch (3 Myr) resulting in regional climate shifts (Ravelo et al., 2004). These changes in climate and sea surface temperature affect the biogenic production rate as noted in the last few million years (e.g., Ruddiman & Janecek, 1989).

4. Conclusion

We determine the sediment thickness along the equatorial MAR inferred from delay times of *P*-to-*S* converted phases at the crust-sediment boundary using teleseismic earthquakes recorded by the PI-LAB seismic network. The average $P_d s$ -*P* delay times range from 0.04 to 0.37 s, which translates to 5- to 82-m thickness across seafloor age from 0.3 to 70.9 Myr old.

Sediment thickness increases with age, as expected. Our thickness estimates are in agreement with the global model for young (< 15–20 Myr old) seafloor but are substantially thinner at older ages (Whittaker et al., 2013). The discrepancy could be influenced by differences in site location; those of the ocean drilling measurements were located in basins, whereas ours are often on topographic highs. Our result may represent a lower limit and suggests the rate of sediment deposition decreases from 5 mm/kyr at about 10 Myr to 1.2 mm/kyr for older seafloor. The higher rate of recent deposition may be caused by changes in climate that caused higher flux of dust and/or biogenic activity.



Acknowledgments

We acknowledge funding from the Natural Environment Research Council (NE/M003507/1 and NE/K010654/1) and the European Research Council (GA 638665). Ocean bottom seismometers were provided by Scripps Institution of Oceanography, Lamont-Doherty Earth Observatory, and the Institut de Physique du Globe de Paris. We would like to thank the crew aboard the ships RV Marcus G Langseth (MGL02-16) and the RRS Discovery (DY072). The OBS data described here are archived at the **IRIS Data Management Center** (http://www.iris.edu) under network code XS 2016-2017. Figures were made using Generic Mapping Tools software (Wessel et al., 2013).

References

- Alibés, B., Canals, M., Alonso, B., Lebreiro, S. M., & Weaver, P. P. E. (1996). Quantification of neogene and quaternary sediment input to the Madeira Abyssal Plain. *Geogaceta*, 20(2), 394–397.
- Alibés, B., Rothwell, R., Canals, M., Weaver, P., & Alonso, B. (1999). Determination of sediment volumes, accumulation rates and turbidite emplacement frequencies on the Madeira Abyssal Plain (NE Atlantic): A correlation between seismic and borehole data. *Marine Geology*, 160(3), 225–250. https://doi.org/10.1016/S0025-3227(99)00026-2
- Bell, S. W., Ruan, Y., & Forsyth, D. W. (2015). Shear velocity structure of abyssal plain sediments in Cascadia. Seismological Research Letters, 86(5), 1247. https://doi.org/10.1785/0220150101
- Bird, P. (2003). An updated digital model of plate boundaries. *Geochemistry, Geophysics, Geosystems*, 4(3), 1027. https://doi.org/10.1029/2001GC000252
- de Menocal, P. B. (1995). Plio-pleistocene African climate. Science, 270(5233), 53-59. https://doi.org/10.1126/science.270.5233.53
- Doran, A. K., & Laske, G. (2017). Ocean bottom seismometer instrument orientations via automated Rayleigh wave arrival angle measurements ocean bottom seismometer instrument orientations via automated Rayleigh wave arrival angle measurements. *Bulletin of the Seismological Society of America*, 107(2), 691. https://doi.org/10.1785/0120160165
- Ewing, J., & Ewing, M. (1967). Sediment distribution on the Mid-Ocean Ridges with respect to spreading of the sea floor. *Science*, 156(3782), 1590–1592. https://doi.org/10.1126/science.156.3782.1590
- Hamilton, E. L. (1976). Shear-wave velocity versus depth in marine sediments: A review. *Geophysics*, 41(5), 985–996. https://doi.org/10.1190/ 1.1440676
- Hamilton, E. L. (1979). Vp/Vs and Poisson's ratios in marine sediments and rocks. The Journal of the Acoustical Society of America, 66(4), 1093–1101. https://doi.org/10.1121/1.383344
- Harmon, N., Forsyth, D. W., Lamm, R., & Webb, S. C. (2007). P and S wave delays beneath intraplate volcanic ridges and gravity lineations near the East Pacific Rise. *Journal of Geophysical Research*, *112*, B03309. https://doi.org/10.1029/2006JB004392
- Harmon, N., Rychert, C. A., Agius, M. R., Tharimena, S., Le Bas, T., Kendall, J. M., & Constable, S. (2018). Marine geophysical investigation of the chain fracture zone in the Equatorial Atlantic from the PI-LAB experiment. *Journal of Geophysical Research: Solid Earth*, 123. https://doi.org/1029.2018JB015982
- Horning, G., Canales, J. P., Carbotte, S. M., Han, S., Carton, H., Nedimovi, M. R., & Keken, P. E. (2016). A 2-D tomographic model of the Juan de Fuca plate from accretion at axial seamount to subduction at the Cascadia margin from an active source ocean bottom seismometer survey. *Journal of Geophysical Research: Solid Earth*, 121, 5859–5879. https://doi.org/10.1002/2016JB013228
- Karson, J., Cannat, M., Miller, D., & Elthon, D. (1997). Proceedings of the Ocean Drilling Program. 153 Scientific Results, Ocean Drilling Program https://doi.org/10.2973/odp.proc.sr.153.1997
- Kelemen, P. B., Kikawa, E., Miller, D. J., & Shipboard Scientific Party (2007). Leg 209 summary: Processes in a 20-km-thick conductive boundary layer beneath the Mid-Atlantic Ridge, 14° – 16° N In P. B. Kelemen, E. Kikawa, & D. J. Miller (Eds.), Proc. ODP, Sci. Results, 209 (pp. 1–33). College Station, TX: Ocean Drilling Program. https://doi.org/10.2973/odp.proc.sr.209.001.2007
- Ku, T.-L., Broecker, W. S., & Opdyke, N. (1968). Comparison of sedimentation rates measured by paleomagnetic and the ionium methods of age determination. *Earth and Planetary Science Letters*, 4(1), 1–16. https://doi.org/10.1016/0012-821X(68)90046-0
- Lewis, B. T. R., & Dorman, L. M. (1998). Recording teleseisms on the seafloor: An example from the Juan de Fuca plate. Bulletin of the Seismological Society of America, 88(1), 107.
- Magotra, N., Ahmed, N., & Chael, E. (1989). Single-station seismic event detection and location. *IEEE Transactions on Geoscience and Remote Sensing*, 27(1), 15–23. https://doi.org/10.1109/36.20270
- Mehouachi, F., & Singh, S. C. (2018). Water-rich sublithospheric melt channel in the equatorial Atlantic Ocean. Nature, 11(1), 65–69. https://doi.org/10.1038/s41561-017-0034-z
- Molnar, P., England, P., & Martinod, J. (1993). Mantle dynamics, uplift of the Tibetan Plateau, and the Indian monsoon. *Reviews of Geophysics*, 31(4), 357–396. https://doi.org/10.1029/93RG02030
- Müller, R. D., Sdrolias, M., Gaina, C., & Roest, W. R. (2008). Age, spreading rates, and spreading asymmetry of the world's ocean crust. Geochemistry, Geophysics, Geosystems, 9, Q04006. https://doi.org/10.1029/2007GC001743
- Nafe, J. E., & Drake, C. L. (1957). Variation with depth in shallow and deep water marine sediments of porosity, density and the velocities of compressional and shear waves. *Geophysics*, 22(3), 523. https://doi.org/10.1190/1.1438386
- Nolet, G., & Dorman, L. M. (1996). Waveform analysis of Scholte modes in ocean sediment layers. *Geophysical Journal International*, 125(2), 385–396. https://doi.org/10.1111/j.1365-246X.1996.tb00006.x
- Ravelo, A. C., Andreasen, D. H., Lyle, M., Lyle, A. O., & Wara, M. W. (2004). Regional climate shifts caused by gradual global cooling in the Pliocene epoch. *Nature*, 429(6989), 263.
- Ritsema, J., Van Heijst, H., Woodhouse, J. H., & Deuss, A. (2009). Long-period body wave traveltimes through the crust: Implication for crustal corrections and seismic tomography. *Geophysical Journal International*, 179(2), 1255–1261. https://doi.org/10.1111/j.1365-246X. 2009.04365.x
- Ruan, Y., Forsyth, D. W., & Bell, S. W. (2014). Marine sediment shear velocity structure from the ratio of displacement to pressure of Rayleigh waves at seafloor. *Journal of Geophysical Research: Solid Earth*, 119, 6357–6371. https://doi.org/10.1002/2014JB011162
- Ruddiman, W., & Janecek, T. (1989). Pliocene-Pleistocene biogenic and terrigenous fluxes at equatorial Atlantic sites 662, 663, and 664, *Proceedings of the Ocean Drilling Program, 108 Scientific Results* (Vol. 108, pp. 211–240): Ocean Drilling Program. https://doi.org/ 10.2973/odp.proc.sr.108.165.1989
- Ruddiman, W. F., Prell, W. L., & Raymo, M. E. (1989). Late Cenozoic uplift in southern Asia and the American West: Rationale for general circulation modeling experiments. *Journal of Geophysical Research*, 94(D15), 18,379–18,391. https://doi.org/10.1029/JD094iD15p18379
- Ruddiman, W., Sarnthein, M., Backman, J., Baldauf, J., Curry, W., Dupont, L., et al. (1989). Late Miocene to Pleistocene evolution of climate in Africa and the low-latitude Atlantic: Overview of Leg 108 results, 108 (pp. 463). https://doi.org/10.2973/odp.proc.sr.108.173.1989 Shipboard Scientific Party (1988a). Proceedings initial reports, 7. Site 662. In W. Ruddiman, M. Sarnthein, J. Baldauf, et al. (Eds.), *Proc. ODP*,
- Shipboard Scientific Party (1988a). Proceedings initial reports, 7. Site 662. In W. Rudolman, M. Sarnthein, J. Baldauf, et al. (Eds.), Proc. ODP, Init. Repts., 108 (pp. 487–556). College Station, TX: Ocean Drilling Program. https://doi.org/10.2973/odp.proc.ir.108.107.1988
 Shipboard Scientific Party (1988b). Site 663. In W. Rudolman, M. Sarnthein, J. Baldauf, et al. (Eds.), Proc. ODP, Init. Repts., 108 (pp. 557–617).
- College Station, TX: Ocean Drilling Program. https://doi.org/10.2973/odp.proc.ir.108.108.1988 Rychert, C. A., Harmon, N., & Tharimena, S. (2018). Scattered wave imaging of the oceanic plate in Cascadia. *Science Advances*, *4*(2),
- Rycherr, C. A., Harmon, N., & Tharimena, S. (2018). Scattered wave imaging of the oceanic plate in Cascadia. Science Advances, 4(2), eaao1908. https://doi.org/10.1126/sciadv.aao1908
- Seher, T., Crawford, W. C., Singh, S. C., Cannat, M., Combier, V., & Dusunur, D. (2010). Crustal velocity structure of the Lucky Strike segment of the Mid-Atlantic Ridge at 37Åř ° N from seismic refraction measurements. *Journal of Geophysical Research*, 115, B03103. https://doi.org/10.1029/2009JB006650

Shearer, P. M., & Orcutt, J. A. (1987). Surface and near-surface effects on seismic waves — Theory and borehole seismometer results. Bulletin of the Seismological Society of America, 77(4), 1168.

- Tiedemann, R., Sarnthein, M., & Shackleton, N. J. (1994). Astronomic timescale for the Pliocene Atlantic δ¹⁸o and dust flux records of ocean drilling program site 659. Paleoceanography, 9(4), 619–638. https://doi.org/10.1029/94PA00208
- Wessel, P., Smith, W. H. F., Scharroo, R., Luis, J., & Wobbe, F. (2013). Generic Mapping Tools: Improved version released. *Eos, Transactions American Geophysical Union*, 94(45), 409–410. https://doi.org/10.1002/2013EO450001
- Whittaker, J. M., Goncharov, A., Williams, S. E., Müller, R. D., & Leitchenkov, G. (2013). Global sediment thickness data set updated for the Australian-Antarctic Southern Ocean. *Geochemistry, Geophysics, Geosystems*, 14, 3297–3305. https://doi.org/10.1002/ggge.20181
- Zhang, Z., Ramstein, G., Schuster, M., Li, C., Contoux, C., & Yan, Q. (2014). Aridification of the Sahara desert caused by Tethys Sea shrinkage during the late Miocene. *Nature*, *513*(7518), 401. https://doi.org/10.1038/nature13705